Author's response to reviews

Title: Endoscopic management of hypertensive intraventricular haemorrhage with obstructive hydrocephalus

Authors:

Yad Ram Yadav (yadavyr@yahoo.co.in)
Gaurav Mukerji (g.mukerji@imperial.ac.uk)
Ravikiran Shenoy (r.shenoy@imperial.ac.uk)
Abhijeet Basoor (drasbasoor@hotmail.com)
Gaurav Jain (gauraviain1000@yahoo.com)
Adam Nelson (drnelsonadam@yahoo.co.uk)

Version: 3 Date: 1 January 2007

Author's response to reviews: see over
To,
The Editor,
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

London, 01 January 2007

Sub: Submission of revised manuscript

Dear Editor,

Please find the revised version of the manuscript with the requested changes. The following revision and formatting changes have been made to the manuscript:

1. The grammatical error in the 3rd Paragraph of discussion has been corrected by adding the word ‘be’ and now reads – ‘…. even if the whole clot may not be suitable for endoscopic removal.’
2. Sentence case has been used in the title.
3. The details of the ethical approval for the study has been added to the Methods section
4. Tables - shading has been removed from the table.
5. Conclusion section has been renamed as ‘Conclusions’
6. 'The authors declare that they have no competing interests' added to the ‘competing interests’ section.
7. We have acknowledged the radiology, operating theatre and nursing staff at NSCB Medical College and Hospital, Jabalpur for their support during the study. None of the authors or the study was externally funded.

We hope that this manuscript now conforms to the Journal’s format requirements and is acceptable for publication.

Yours sincerely,

GAURAV MUKERJI